Author's response to reviews

Title: Giant aneurysm of the left main coronary artery with fistulous communication to the right atrium

Authors:

Zhicheng Zhu (zhu5wolf@hotmail.com)
Rihao Xu (xu6mouse@hotmail.com)
Dan Li (lidan99@jlu.edu.cn)
Yong Wang (wangyong.hahaha@163.com)
Bo Li (adam1911@21cn.com)
Tiance Wang (w19670721@163.com)
Shudong Zhang (zsd_888905@sina.com)
Kexiang Liu (kxliu64@hotmail.com)

Version: 4 Date: 14 June 2015

Author's response to reviews: see over
Dear editor Journal of Cardiothoracic Surgery

My manuscript, giant aneurysm of the left main coronary artery with fistulous communication to the right atrium, was revised according to the reviewers' comments, and the itemized response to each reviewer's comments is attached. Many thanks for your suggestion. Thanks very much again for your attention to our paper. Once again, thank you for your help to our paper processing.

The authors are:
Zhicheng Zhu; Yong Wang; Rihao Xu; Dan Li, Tiance Wang; Bo Li; Shudong Zhang; Kexiang Liu

Yours sincerely,
Kexiang Liu

Corresponding to: Kexiang Liu, Ph.D. Department of Cardiovascular Surgery, Second Hospital of Jilin University, Jilin University, Changchun, Jilin 130041

Tel.: +86 0431 88796791

E-mail addresses: kxliu64@hotmail.com

Author's response to reviews

Reviewer: KASRA SHAIKHREZAI

Major compulsory revisions
1. There are some minor syntax problems in the newly added phrases which need to be addressed.

   Answer: Thank you for your comment.

   Change: The newly added phrases have been addressed by the help of professionals in the green color.

Your comment and advice is important for us, which would make our study better. At last, I want to thank you sincerely for your suggestions and your time for reviewing the paper.